Author's response to reviews

Title: Opt-out as an acceptable method of obtaining consent in medical research: a short report.

Authors:

Akke Vellinga (akke.vellinga@nuigalway.ie)
Martin Cormican (martin.cormican@hse.ie)
Belinda Hanahoe (belinda.hanahoe@hse.ie)
Kathleen Bennett (bennettk@tcd.ie)
Andrew W Murphy (andrew.murphy@nuigalway.ie)

Version: 4 Date: 29 March 2011

Author's response to reviews:

Dear BMC editorial team,

Please find attached the manuscript with the requested revisions. As it is a short paper, no distinction was made between the discussion and the conclusions. All other revisions have been done.

Thank you,

Akke Vellinga